Author's response to reviews

Title: Emergency room (ER) visits of children under age one: effects of immigrant status and maternal education level. A cohort study in the province of Reggio Emilia (Italy)

Authors:

paola ballotari (paola.ballotari@ausl.re.it)
stefania d'angelo (sd@mrc.soton.uk)
laura bonvicini (laura.bonvicini@ausl.re.it)
serena broccoli (serena.broccoli@ausl.re.it)
nicola caranci (ncaranci@regione.emilia-romagna.it)
silvia candela (silvia.candela@gmail.com)
paolo giorgi rossi (paolo.giorgirossi@ausl.re.it)

Version: 3 Date: 28 December 2012

Author's response to reviews:

I have included in the methods section the details about availability and permission of data use